Author's response to reviews

Title: Evaluation of an online Diabetes Needs Assessment Tool (DNAT) for health professionals: a randomised controlled trial

Authors:

Sara Schroter (sschroter@bmj.com)
Dean Jenkins (djenkins@bmjgroup.com)
Rebecca Playle (playlera@cardiff.ac.uk)
Kieran Walsh (KMWalsh@bmjgroup.com)
Courtenay Probert (cprobert@bmjgroup.com)
Thomas Kellner (thomas_kellner@msd.com)
Gerhard Arnhofer (gerhard_arnhofer@merck.com)
David Owens (owensdr@Cardiff.ac.uk)

Version: 4 Date: 29 July 2009

Author's response to reviews:

Dear Editor

We have made the following three revisions to the manuscript to reflect changes in the protocol since writing the submitted versions

P10: Assessments and procedures
We have changed the text to indicate that learners will receive 6 reminders to use the learning materials

P10: Assessments and procedures
We have changed the text to indicate that learners will receive a series of reminders to complete the test and survey at follow-up

P21: Funding
We have changed the text to
“All other expenses for the study, except translation costs, were paid by BMJ Group”.

Kind regards

Sara